Author's response to reviews

Title: An observational study of nutrition and physical activity behaviours, knowledge and advice in pregnancy: Are we doing enough?

Authors:

Susan J de Jersey Ms (susie.dejersey@qut.edu.au)
Jan M Nicholson Prof (JNicholson@parentingrc.org.au)
Leonie K Callaway A/Prof (l.callaway@uq.edu.au)
Lynne A Daniels Prof (l2.daniels@qut.edu.au)

Version: 2 Date: 6 December 2012

Author's response to reviews:

Dear Editor,

The revisions to this manuscript include:
Expansion of ethics statement to include- Specification of ethics committees including approval numbers, participants provided written informed consent and the research was conducted in accordance with ethical standards for human experimentation.
Statement of competing interests from authors.

Thank you for alerting me to these issues.

Kind Regards

Susie de Jersey